Author’s response to reviews

Title: Risk factors for nasopharyngeal carriage of drug-resistant Streptococcus pneumoniae: data from a nation-wide surveillance study in Greece.

Authors:

Ioannis I Katsarolis (ikatsarolis@yahoo.com)
Garyphalia G Poulakou (gpoul@med.uoa.gr)
Antonios A Analitis (aanalit@med.uoa.gr)
Irini I Matthaiopoulou (matthaiopoulou@hotmail.com)
Emmanuel E Rolides (rolides@med.auth.gr)
Charalampos C Antachopoulos (rolides@med.auth.gr)
Dimitrios DA Kafetzis (kafetzis@ath.forthnet.gr)
Georgios GL Daikos (gdaikos@med.uoa.gr)
Regina R Vorou (vorou@keelpno.gr)
Christina C Koubaniou (panzois@yahoo.gr)
Ioannis I Pneumatikos (ipnevmat@med.duth.gr)
Georgios G Samonis (georgsec@med.uoc.gr)
Vasiliki V Syriopoulou (vsyriop@med.uoa.gr)
Helen H Giamarellou (hgiama@ath.forthnet.gr)
Kyriaki K Kanellakopoulou (kyrkanel@yahoo.gr)

Version: 3 Date: 13 March 2009

Author’s response to reviews: see over
Dear Editor,

Please let me first thank you on behalf of the authors for accepting our manuscript for publication in BMC Infectious Diseases.

Please find attached the formatted revised manuscript entitled “Risk factors for nasopharyngeal carriage of drug-resistant *Streptococcus pneumoniae*: data from a nation-wide surveillance study in Greece” by I Katsarolis et al.

Formatting changes have been made according to your email instructions (both major and minor).

**Major revisions**

- Author list - The list of authors in the manuscript has been written exactly as they are in the submission system.
- References - The reference list contains all authors names.
- Tables - No vertical lines in the tables.
- Table size – Formatted to fit into an A4 portrait page.
- Table shading – No shading and colours within the tables.
- Figure title - The image file does not include the title or figure number.

**Minor revisions**

- Line numbers – Removed
- Highlighting/tracking – Removed
- Abstract - The Abstract is present in manuscript and matches exactly what is in the submission system.
- Transparency Declarations-Competing Interests -> Competing interests
- Underlining - No underlining
- Figure cropping – Figure cropped appropriately

In case of any problem, do not hesitate to contact me

Yours sincerely,

Dr Ioannis Katsarolis